Author's response to reviews

Title: Influence of Ketotifen, Cromolyn Sodium, and Compound 48/80 on the survival rates after intestinal ischemia reperfusion injury in rats

Authors:

Hei Zi Qing (heiziqing0530@hotmail.com)
Gan Xiao Liang (ganxiaoliang@21cn.com)
Huang Pin Jie (hpjie@126.com)
Wei Jing (weijing@163.com)
Shen Ning (paulshenyh@yahoo.com)
Gao Wan Ling (wmrgwl@21cn.com)

Version: 5 Date: 20 August 2008

Author's response to reviews: see over
**Author's response to reviews**

Title: Influence of Ketotifen, Cromolyn Sodium, and Compound 48/80 on the survival rates after intestinal ischemia reperfusion injury in rats

Authors:

Zi-qing Hei- [heiziqing0530@hotmail.com](mailto:heiziqing0530@hotmail.com);

Xiao-liang Gan- [ganxiaoliang@21cn.com](mailto:ganxiaoliang@21cn.com);

Pin-jie Huang- [hpjie@126.com](mailto:hpjie@126.com);

Jing Wei- [weijing@163.com](mailto:weijing@163.com);

Ning Shen- [paulshenyh@yahoo.com.cn](mailto:paulshenyh@yahoo.com.cn);

Wan-ling Gao- [wmrgwl@21cn.com](mailto:wmrgwl@21cn.com)
However before acceptance, we strongly encourage you to include an Acknowledgements section between the Authors’ contributions section and Reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include their source(s) of funding. Please also acknowledge anyone who contributed materials essential for the study.

The section of Authors’ contributions was replaced by the following paragraph:

Authors' contributions
H-ZQ participated in the conception and design of the study, and revised the manuscript. G-XL established the experimental setup, and drafted the manuscript. H-PJ and W-J carried out biochemical study and carried out pathological examination. S-N and G-WL collected and analyzed the data. All authors read and approved the final manuscript.

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

Please list the source(s) of funding for the study, for each author, and for the manuscript preparation in the acknowledgements section. Authors must describe the role of the funding body, if any, in study design; in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

We added the section of Acknowledgements, however, there was no funding supported the experiment.

Acknowledgements
We would like to thank Jia-jun Liu for his help in the experiment.

You should also carefully copyedit your manuscript one last time to ensure all grammatical and spelling errors are corrected.

Done